% LaTeX2e file in llncs formal with TOC and indexes for TEST 2007.
% You can store this page to file test2007.tex
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% In order to generate an Author Index do the following:
% After TeXing this document start the program MakeIndex by typing
%   makeindex -s sprmindx.sty <filename>
% into the command line.
% (On DOS systems you may need to use the command MAKEINDX.)
% Now TeX this file once again, then you will get an Author Index.
% TeX this file once more, then the TOC will be complete.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentclass{llncs}
%
\usepackage{makeidx}  % allows for index generation
\makeindex
%
\begin{document}
%
\frontmatter          % for the preliminaries
%
\pagestyle{headings}  % switches on printing of running heads
%
\chapter*{Preface}
%
TEST 2007 was held at...
%\input{preface.tex}

\begin{flushright}\noindent
January 2001\hfill Your Name Here\\
TEST 2007 Program Chair
\end{flushright}
%
\chapter*{External reviewers}
%
\begin{multicols}{3}
\noindent
Some One\\ 
\end{multicols}
%
\chapter*{{\huge TEST 2007}\\
\ \\
July 2007 Test}

\vspace{-\bigskipamount}
\vspace{-\bigskipamount}
\begin{center}
{\large Some Location, Some City, State or Country\\
Dates}

\bigskip\bigskip Sponsored by \emph{Somebody}

\medskip \mbox{Organized in cooperation with \emph{Somebody else}}

\bigskip\bigskip\textbf{\large General Chair}

 \smallskip
Name and Affiliation

\bigskip\bigskip\textbf{\large Program Chair}

\smallskip
Name and Affiliation


\bigskip\textbf{\large Program Commitee}

\smallskip\begin{tabular}{@{}p{4.2cm}@{}p{7.2cm}@{}}
David Beckham	& affiliations\\ 
Fabio Cannavaro	& affiliations\\ 
Angelos Charisteas	& affiliations\\ 
Joy Fawcett	& affiliations\\ 
Danielle Fotopoulos	& affiliations\\ 
Julie Foudy	& affiliations\\ 
Stylianos Giannakopoulos	& affiliations\\ 
Mia Hamm	& affiliations\\ 
Devvyn Hawkins	& affiliations\\ 
Angela Hucles	& affiliations\\ 
Jena Kluegel	& affiliations\\ 
Amy LePeilbet	& affiliations\\ 
Paolo Maldini	& affiliations\\ 
Antonios Nikopolidis	& affiliations\\ 
Michael Owen	& affiliations\\ 
Francesco Totti	& affiliations\\ 
Zinedine Zidane	& affiliations\\ 
\end{tabular}

%\bigskip\bigskip\textbf{\large Organizing Committee}
%
%\smallskip\begin{tabular}{@{}p{4.2cm}@{}p{7.2cm}@{}}
%One Person    & Affiliations \\
%Second Person & Affiliations \\
%\end{tabular}
\end{center}

%
%\section*{Sponsoring Institutions}
%
%Bernauer-Budiman Inc., Reading, Mass.\\
%The Hofmann-International Company, San Louis Obispo, Cal.\\
%Kramer Industries, Heidelberg, Germany
%
\tableofcontents
%
\mainmatter              % start of the contributions

%% add session names to the table-of-contents with the \addtocmark command
%\addtocmark{Zero-Knowledge}

\setcounter{page}{1}
\title{Cryptanalysis of two identification schemes based on an ID-based cryptosystem}
\author{Qiang Tang \and  Chris J. Mitchell}
\index{Tang, Qiang}
\index{Mitchell, Chris J.}
\maketitle
\clearpage

\setcounter{page}{18}
\title{Tag-KEM/DEM: A New Framework for Hybrid Encryption}
\author{Masayuki ABE \and  Rosario Gennaro \and  Kaoru Kurosawa}
\index{ABE, Masayuki}
\index{Gennaro, Rosario}
\index{Kurosawa, Kaoru}
\maketitle
\clearpage

\setcounter{page}{40}
\title{Improved Proxy Re-Encryption Schemes with Applications to Secure Distributed Storage}
\author{Giuseppe Ateniese \and  Kevin Fu \and  Matthew Green \and  Susan Hohenberger}
\index{Ateniese, Giuseppe}
\index{Fu, Kevin}
\index{Green, Matthew}
\index{Hohenberger, Susan}
\maketitle
\clearpage

\setcounter{page}{60}
\title{Adversarial Model for Radio Frequency Identification}
\author{Gildas Avoine}
\index{Avoine, Gildas}
\maketitle
\clearpage

\setcounter{page}{77}
\title{Mixing properties of triangular feedback shift registers}
\author{Bernd Schomburg}
\index{Schomburg, Bernd}
\maketitle
\clearpage

\setcounter{page}{94}
\title{A sufficient condition for key-privacy}
\author{Shai Halevi}
\index{Halevi, Shai}
\maketitle
\clearpage

\setcounter{page}{115}
\title{Update on SHA-1}
\author{Vincent Rijmen \and  Elisabeth Oswald}
\index{Rijmen, Vincent}
\index{Oswald, Elisabeth}
\maketitle
\clearpage

\setcounter{page}{131}
\title{Collusion Resistant Broadcast Encryption With Short Ciphertexts and Private Keys}
\author{Dan Boneh \and  Craig Gentry  \and  Brent Waters}
\index{Boneh, Dan}
\index{Gentry, Craig}
\index{Waters, Brent}
\maketitle
\clearpage

\setcounter{page}{148}
\title{Side Channel Attacks on Implementations of Curve-Based Cryptographic Primitives}
\author{Roberto M. Avanzi}
\index{Avanzi, Roberto M.}
\maketitle
\clearpage

\setcounter{page}{161}
\title{A model and architecture for pseudo-random generation with applications to /dev/random}
\author{Boaz Barak \and  Shai Halevi}
\index{Barak, Boaz}
\index{Halevi, Shai}
\maketitle
\clearpage

\setcounter{page}{180}
\title{Weak keys of pairing based Diffie Hellman schemes on elliptic curves}
\author{A. A. Kalele \and  V. R. Sule}
\index{Kalele, A. A.}
\index{Sule, V. R.}
\maketitle
\clearpage

\setcounter{page}{200}
\title{The Vector Decomposition Problem for Elliptic and Hyperelliptic Curves}
\author{Iwan Duursma \and  Negar Kiyavash}
\index{Duursma, Iwan}
\index{Kiyavash, Negar}
\maketitle
\clearpage

%---------------------------------------------------------------
\setcounter{page}{210}
\addtocmark[2]{Author Index}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\renewcommand{\indexname}{Author Index}
\printindex

\end{document}
